Author's response to reviews

Title: A Novel Case of Diffuse Large B-cell Lymphoma Masked by Epstein Barr Virus Driven Hemophagocytic Lymphohistiocytosis: A case report

Authors:

Sadaf Altaf (altaf.sadaf@mayo.edu)
Grace Arteaga (grace.arteaga@mayo.edu)
Avni Joshi (joshi.avni@mayo.edu)
Vilmarie Rodriguez (rodriguez.vilmarie@mayo.edu)

Version: 2 Date: 28 November 2011

Author's response to reviews:

1- I have updated the title per format of the journal.

2- Informed consent was obtained from the patient as the patient recently turned 18 and is currently a consenting adult.